Author's response to reviews

Title: E-selectin gene polymorphisms are associated with essential hypertension: a case-control pilot study in a Chinese population

Authors:

Zuoguang Wang (wzg70@hotmail.com)
Ya Liu (49924256@qq.com)
Jieling Liu (849267265@qq.com)
Kuo Liu (zoulo567@sohu.com)
Yuqin Lou (645073501@qq.com)
Jie Wen (1204098281@qq.com)
Qiuli Niu (349094431@qq.com)
Shaojun Wen (wzg70@yahoo.com)
Zhaosu Wu (wuzhaosu@263.net)

Version: 13 Date: 25 August 2010

Author's response to reviews:

wzg70@hotmail.com
25th August, 2010

BMC Medical Genetics
Editorial Office,

Dear Editor

We are now resubmitting a revised manuscript by Zuoguang.Wang, et al, entitled “(MS: 1617310499373018) E-selectin gene polymorphisms are associated with essential hypertension: a case-control pilot study in a Chinese population” for the publication in BMC Medical Genetics.

Following the suggestion of editorial office, we have copyedited the paper again. We think it is now ready for publication.

Yours, sincerely
Shaojun.Wen,M.D.